Author's response to reviews

Title: Setting up a surveillance system for sexually transmitted diseases in the general population with prospective data collection from private-practice and public-practice doctors in Hong Kong

Authors:

Joseph TF Lau (jlau@cuhk.edu.hk)
Chunqing Lin (lincq@cuhk.edu.hk)
King Man Ho (eo_shs@dh.gov.hk)
Man Chun Lau (mason-lau@cuhk.edu.hk)
Hi Yi Tsui (hytsui@cuhk.edu.hk)
Jing Gu (gujing@cuhk.edu.hk)
Kuen Kong Lo (eo_shs@dh.gov.hk)

Version: 6 Date: 14 April 2011

Author's response to reviews:

Dear Editor,

Thank you for your email dated Apr 13, 2011. Changes have been made accordingly.

Please find below for details.

1. The list of authors in the manuscript and the submission system is identical now.

2. We have proofread the manuscript for typography.

Thank you again for accepting our manuscript.